

Citation: Priest JR, Osoegawa K, Mohammed N, Nanda V, Kundu R, Schultz K, et al. (2016) *De Novo* and Rare Variants at Multiple Loci Support the Oligogenic Origins of Atrioventricular Septal Heart Defects. PLoS Genet 12(4): e1005963. doi:10.1371/ journal.pgen.1005963

Editor: Kai Jiao, University of Alabama at Birmingham, UNITED STATES

Received: May 12, 2015

Accepted: March 7, 2016

Published: April 8, 2016

Copyright: © 2016 Priest et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Exome data can be found at the dbGAP repository maintained by NIH for researchers who meet the criteria for access to confidential data. Application procedures for dbGAP can be found at (https://dbgap.ncbi.nlm.nih.gov/aa/ wga.cgi?page=login). Exome data from the Toronto single samples is freely available at the EGA repository for researchers who meet the criteria for access to confidential data. Application procedures for EGA cat be found at (https://www.ebi.ac.uk/ega/ about/access) Sage gene expression data from mouse organ development can be found at http:// **RESEARCH ARTICLE**

De Novo and Rare Variants at Multiple Loci Support the Oligogenic Origins of Atrioventricular Septal Heart Defects

James R. Priest^{1,2}, Kazutoyo Osoegawa³, Nebil Mohammed⁴, Vivek Nanda⁵, Ramendra Kundu⁶, Kathleen Schultz⁴, Edward J. Lammer^{4†}, Santhosh Girirajan⁷, Todd Scheetz⁸, Daryl Waggott⁶, Francois Haddad^{2,6}, Sushma Reddy^{1,2}, Daniel Bernstein^{1,2}, Trudy Burns⁹, Jeffrey D. Steimle¹⁰, Xinan H. Yang¹⁰, Ivan P. Moskowitz¹⁰, Matthew Hurles¹¹, Richard P. Lifton^{12,13}, Debbie Nickerson¹⁴, Michael Bamshad^{14,15}, Evan E. Eichler^{13,14}, Seema Mital¹⁶, Val Sheffield^{13,17}, Thomas Quertermous^{2,6}, Bruce D. Gelb¹⁸, Michael Portman¹⁵, Euan A. Ashley^{2,6}*

1 Division of Pediatric Cardiology, Stanford University School of Medicine, Stanford University, Stanford, California, United States of America, 2 Cardiovascular Institute, Stanford University School of Medicine, Stanford University, Stanford, California, United States of America, 3 Department of Pathology, Stanford University School of Medicine, Stanford University, Stanford, California, United States of America, 4 University of California San Francisco Benioff Children's Hospital Oakland, University of California San Francisco, San Francisco, California, United States of America, 5 Department of Vascular Surgery, Stanford University School of Medicine, Stanford University, Stanford, California, United States of America, 6 Division of Cardiovascular Medicine, Stanford University School of Medicine, Stanford University, Stanford, California, United States of America, 7 Departments of Biochemistry, Molecular Biology, and Anthropology, Pennsylvania State University, University Park, Pennsylvania, United States of America, 8 College of Engineering, University of Iowa, Iowa City, Iowa, United States of America, 9 College of Public Health, University of Iowa, Iowa City, Iowa, United States of America, 10 Department of Pathology, University of Chicago, Chicago, Illinois, United States of America, 11 Wellcome Trust Sanger Institute, Hinxton, Cambridge, United Kingdom, 12 Department of Genetics, Yale University, New Haven, Connecticut, United States of America, 13 Howard Hughes Medical Institute, Chevy Chase, Maryland, United States of America, 14 Department of Genome Sciences, University of Washington, Seattle, Washington, United States of America, 15 Department of Pediatrics, University of Washington, Seattle, Washington, United States of America, 16 Department of Pediatrics, University of Toronto, Toronto, Ontario, Canada, 17 Division of Medical Genetics, University of Iowa Carver College of Medicine, Iowa City, Iowa, United States of America, 18 Mindich Child Health and Development Institute, Icahn School of Medicine at Mt. Sinai, New York, New York, United States of America

† Deceased.

* <u>euan@stanford.edu</u>

Abstract

Congenital heart disease (CHD) has a complex genetic etiology, and recent studies suggest that high penetrance *de novo* mutations may account for only a small fraction of disease. In a multi-institutional cohort surveyed by exome sequencing, combining analysis of 987 individuals (discovery cohort of 59 affected trios and 59 control trios, and a replication cohort of 100 affected singletons and 533 unaffected singletons) we observe variation at novel and known loci related to a specific cardiac malformation the atrioventricular septal defect (AVSD). In a primary analysis, by combining developmental coexpression networks with inheritance modeling, we identify a *de novo* mutation in the DNA binding domain of *NR1D2* (p.R175W). We show that p.R175W changes the transcriptional activity of Nr1d2 using an *in vitro* transactivation model in HUVEC cells. Finally, we demonstrate previously

www.mouseatlas.org/. The RNAseq data from the second heart field and DMP generated by the Moskowitz lab has been submitted to GEO (GSE75077).

PLOS GENETICS

Funding: JRP was supported by the Pediatric Scientist Development Program (NIH-NICHD K12-HD000850). The sequencing was supported by an in kind grant to EAA (NIH-NHLBI X01HL115206-01), and SAP corporation (Walldorf, Germany) which played no role in any portion of the study design, recruitment, data collection, or analysis. RPL is supported by Yale Center for Medelian Disorders (U54HG006504-04), and MB and DN are supported by the UW Center for Medelian Genomics (U54HG006493-04). IPM was supported by the NHLBI (R01 HL092153). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: EAA is a co-founder of Personalis Inc. which offers research and clinical sequencing studies similar to those described in this manuscript. unrecognized cardiovascular malformations in the *Nr1d2*^{tm1-Dgen} knockout mouse. In secondary analyses we map genetic variation to protein-interaction networks suggesting a role for two collagen genes in AVSD, which we corroborate by burden testing in a second replication cohort of 100 AVSDs and 533 controls (p = 8.37e-08). Finally, we apply a rare-disease inheritance model to identify variation in genes previously associated with CHD (*ZFPM2*, *NSD1*, *NOTCH1*, *VCAN*, and *MYH6*), cardiac malformations in mouse models (*ADAM17*, *CHRD*, *IFT140*, *PTPRJ*, *RYR1* and *ATE1*), and hypomorphic alleles of genes causing syndromic CHD (*EHMT1*, *SRCAP*, *BBS2*, *NOTCH2*, and *KMT2D*) in 14 of 59 trios, greatly exceeding variation in control trios without CHD (p = 9.60e-06). In total, 32% of trios carried at least one putatively disease-associated variant across 19 loci,suggesting that inherited and *de novo* variation across a heterogeneous group of loci may contribute to disease risk.

Author Summary

Congenital heart disease (CHD) is a leading cause of childhood morbidity in the developed world. There are few prevalent clinical risk factors and though it is possible that up to 90% of risk for CHD may be genetic, the number of genes clinically associated with disease is small. Rather than grouping disparate CHD phenotypes as other studies have done, we studied a single specific malformation- the atrioventricular septal defect (AVSD). Instead of recurrent variation in a handful of genes, we observed *de novo* and inherited variation in 19 genes associated with human disease, syndromic loci, and genes implicated in cardiac development by mouse knockout. The number of loci identified support the longstanding hypothesis of a complex oligogenic inheritance for a single malformation and suggest that analyses of CHD data to include inherited variation may uncover additional loci contributing risk for cardiac malformations.

Introduction

Congenital heart disease (CHD) is the most common congenital malformation and the most common cause of mortality during the first year of life in the United States [1,2]. Most cases occur sporadically without a strong family history or identifiable genetic syndrome, and the primary heritable basis of most non-syndromic congenital heart disease has yet to be identified [3,4]. Studies of affected kindreds and syndromic disease have revealed high-penetrance mutations at a small number of key loci [5]. Exome sequencing and studies of structural variation of mixed cardiac phenotypes focusing on *de novo* events have identified novel disease loci in 4–10% of participants [6,7]. However the remaining majority of non-syndromic subjects in exome and CNV studies are without an identified genetic cause.

Atrioventricular septal defects (AVSD) are a rare cardiac malformation associated to date with a handful of canonical genes in cardiac development (*NKX2-5*, *GATA4*, *GATA6*, *CRELD1*) and may co-occur with certain rare syndromes. A recent study discovered causal variation in the nuclear receptor *NR2F2* in 4% of 125 subjects with AVSD, pinpointing a single additional disease-associated gene [7]. However the 4–10% discovery rate in studies of CHD highlight the observation that for any individual gene, highly penetrant *de novo* coding mutations may only account for a small portion of disease incidence, a phenomenon similar to the

sporadic occurrence and complex genetics of neurodevelopmental disorders [8]. Therefore, expanding the scope of analysis in studies of CHD to include both inherited and *de novo* variation in multiple genes could increase the sensitivity of genetic studies of this heterogeneous group of oligogenic diseases [9-13].

To this end we assembled a multi-institutional cohort combining a discovery cohort of 59 trios with non-syndromic AVSD and a replication cohort of 100 single affected individuals and performed a genetic survey by exome sequencing and array-CGH. In a primary analysis we identified a novel candidate gene for AVSDs using inheritance modeling and prior knowledge of early cardiac gene expression (Fig_1). In secondary analyses, we searched protein interaction networks to identify the contribution of additional loci to this rare cardiovascular malformation. Finally we explored the contribution of rare inherited variation in genes related to other types of human and mouse cardiac malformations to AVSD.

Results

The RTG Pipeline Yields a Highly Sensitive and Accurate Variant Call Set

We determined the sensitivity of our informatics approach using a recently described consensus standard dataset of exonic 24,734 variants from NA12878 [14]. Raw exome data from a well-characterized individual was analyzed with BWA/GATK 3.2 best practices and the RTG version 3.3.2 software pipeline (Real Time Genomics Inc., Hamilton, New Zealand). RTG displayed greater sensitivity detecting 84.5% consensus standard variants compared to 79.9% for BWA/GATK (S1 Table). Using the RTG pipeline we analyzed exome sequencing data on 159 individuals with AVSD but without a syndrome (a discovery cohort of 59 trios, and a replication cohort of 100 singletons derived largely from a published study of AVSDs (S2 Table), along with 710 controls without congenital heart disease (59 trios, 533 singletons). The affected patients were situs solitus with a simple AVSD. Patients with other cardiac defects, heterotaxy, anatomical malformations, or developmental delay were excluded. Across all individuals, called variants displayed a median Ti/Tv ratio of 3.10, and a median of 89.6% phased genotypes within trio probands, which suggested an overall highly sensitive and accurate variant call set (S3 Table). All *de novo* variants and insertions/deletions of interest were confirmed by Sanger sequencing.

A Rare-Disease Inheritance Model and Developmental Coexpression Networks Identify *NR1D2*

Within the 59 AVSD trios and 59 control trios we analyzed variants with minor allele frequency of 0.03 or less in the 61,468 multiethnic individuals in the EXaC consortium [15]. With the remaining protein-altering variants, we applied a rare-disease inheritance model to select only variants displaying classical inheritance patterns associated with sporadic presentation of a rare disease (*de novo*, homozygous, compound heterozygous) (Fig 1) [16]; this filtering process yielded a list of 710 variants in 399 genes in the 59 AVSD trios (S4 Table).

To identify novel genes involved in cardiac development and disease among the 399 loci, we reanalyzed 72 digital gene expression datasets derived from 22 tissues during mouse embryonic development (www.mouseatlas.org) [17] (S1 Fig). The tissue types included the AV-canal along with 5 other cardiac tissues, and 16 other tissues from other organs or structures. After constructing co-expression modules using unsupervised weighted-gene coexpression network analysis, we observed that one of the co-expression modules expressed in mouse AV-canal tissue included four of six genes known to cause AVSDs (*GATA4, GATA6, NKX2-5*, and





Fig 1. Analytical approach for disease gene discovery in a cohort of AVSD trios. Protein altering variants meeting allele frequency, quality, and read depth cutoffs from 59 trios were sorted with an inheritance model consistent with rare disease which combines *de novo*, compound heterozygous, and rare homozygous variants to collate variants into a list of genes [16]. Three analytical approaches were applied to the final dataset. Primary Analysis. *De novo* variants in the AVSD trios were cross-referenced with the genes

in a module highly enriched for CHD and cardiac development obtained from unsupervised weighted-gene coexpression network analysis to identify a novel AVSD gene. Secondary Analyses. Variants from the AVSD trios were mapped onto a protein interaction network followed by burden testing in a replication cohort (details in Fig 3). In a final analysis employing inheritance modeling, *de novo*, compound heterozygous, and rare homozygous loci observed in the AVSD trios were compared with a predefined list of genes associated with human or mouse cardiac malformations. Statistical comparisons were performed in a control group of 59 control trios without CHD.

doi:10.1371/journal.pgen.1005963.g001

CRELD1, p = 6.57e-05, one-tail hypergeometric test) along with 69 of 756 genes related to other human or mouse cardiac malformations (<u>S5 Table</u>) (p = 7.81e-08, one-tail hypergeometric test). These observations suggested the discovery of a co-expression module highly enriched for genes related to heart development and cardiac malformations.

Intersecting the genes in this unique coexpression module (S6 Table) with the 399 genes identified by the rare-disease inheritance model, two probands from the 59 AVSD trios displayed *de novo* mutations. One proband had a missense mutation in a non-conserved residue of *KCNJ3* and a second proband had a missense variant in *NR1D2*. The *KCNJ3* variant has been observed in low frequencies in European and African populations in the ExAC database and the available literature on *Kcnj3* knockout animals did not suggest occult cardiovascular malformations [18,19]. By contrast, the *NR1D2* variant causes an arginine to tryptophan mutation at position 175 (p.R175W) in a highly conserved DNA binding domain (Fig 2a and 2b). *NR1D2* is a transcriptional co-repressor and modulator without a described role in cardiac development [20]. *De novo* mutations in *NR1D2* or any gene in the cardiac malformation module were absent from 59 control trios without congenital heart disease. Among the 61,468 putatively healthy individuals cataloged in the ExAC database there was only a single non-synonymous mutation in the surrounding five protein residues surrounding the p.R175W allele. Overall the data were suggestive that this *de novo* allele might impact the function of *NR1D2*.

The p.R175W Mutation Impacts Transcriptional Activity of Nr1d2 in vitro

To characterize the transcriptional activity of the p.R175W mutation we developed an *in vitro* transactivation assay performed in HUVEC cells, which employed a murine Nr1d2 expression vector co-transfected with an Nr1d2 response element (RE) consisting of 5 tandem repeats of a conserved binding site upstream of a minimal CMV promoter driving GFP. As NR1D2 is thought to act as a transcriptional co-repressor, a positive change in transcriptional activity of the RE vector may represent a decrease in the transcriptional co-repressor activity of Nr1d2. In this *in vitro* assay, the p.R175W mutation displayed an increased transcriptional activity relative to the wild-type allele (Fig 2c), which suggested that the p.R175W mutation in a conserved region of the DNA binding domain might functionally impair the native co-repressor function of NR1D2.

The *Nr1d2*^{tm1-Dgen} -/- Mouse Shows AVSDs and Related Cardiovascular Malformations

Though reports of a previous characterization of an *Nr1d2* knockout allele did not show cardiovascular malformations [20], a percentage of homozygous knockout animals have been reported to die within hours of birth consistent with the presence of hemodynamically significant cardiac malformations [21]. Two pairs of heterozygous founder animals were bred, yielding 17 pups (2 wild type, 7 *Nr1d2*^{tm1-Dgen} +/-, and 8 *Nr1d2*^{tm1-Dgen} -/-) which did not deviate obviously from expected Mendelian allelic ratios. Upon careful histological examination of two





Fig 2. Mutation at a highly conserved arginine residue in the DNA binding domain of *NR1D2* is associated with AVSD. (a) Multiple alignment of amino acid residues in the C-terminal aspect of the DNA binding domain of NR1D2 show strong conservation through evolution to the *Drosophila* paralog Eip75B. (b) A 3d representation of the crystal structure of NR1D1 complexed to DNA. The NR1D1 and NR1D2 DNA binding domains display 96% sequence identity (S3 Fig), and the altered arginine residue (black arrow) contacts the minor groove of DNA in the C-terminal portion of the DNA binding domain (RCSB 1A6Y) [82]. (c) The p.R175W mutation shows increased transcriptional activity in a cell culture assay (p = 0.0022, two-tailed t-test). A wild-type or mutant *Nr1d2* construct was transfected in HUVEC cells along with a vector containing a synthetic response element driving a GFP reporter. Mean and standard deviation for each condition are displayed. (d) Two representative cardiac lesions in *Nr1d2^{tm1-Dgen}* -/- mouse hearts. An oblique coronal section of the heart of a spontaneously deceased P0 *Nr1d2^{tm1-Dgen}* -/- pup reveals a primum ASD and suggests a common AV-valve (white arrow) indicative of AVSD (star). A coronal section of an e17.5 *Nr1d2^{tm1-Dgen}* -/- heart shows an inlet VSD (plus sign) which is a type of VSD closely related to AVSDs in cardiac development [78,83]. Abbreviations: LV-left and RV-right ventricles, RA-right atrium, IVS-interventricular septum, VSD-ventricular septal defect, ASD-atrial septal defect.

doi:10.1371/journal.pgen.1005963.g002

spontaneously deceased $Nr1d2^{tm1-Dgen}$ -/- pups at P0 we detected a previously undescribed AVSD phenocopy (Fig 2d). We performed additional matings of +/- and -/- animals, and sacrificed mothers to obtain embryos at e16.5 and e17.5. A single -/- animal at e16.5 displayed an AVSD and a single -/- animal at e17.5 displayed an inlet ventricular septal defect which is closely related to AVSDs (Fig 2d). In total, 4 out of 15 -/- hearts assayed displayed a cardiac defect. Thus by combining inheritance modeling with a gene-coexpression network enriched for genes causing CHD, we identified a variant in *NR1D2* which impacts transcriptional activity *in vitro*, and uncovered previously unrecognized cardiovascular malformations in an *Nr1d2* knockout allele. Together these data suggest a new role for the transcriptional repressor *NR1D2* in cardiac development and human disease.

Protein Interaction Networks Identified in the AVSD-trios are Expressed in Cardiac Development

The relatively low discovery rate in rare-variant association studies of CHD suggests that alternative analytical approaches may be necessary to distinguish the contribution of novel loci to disease risk [6,7]. Protein interaction networks have successfully integrated known disease genes to discover the impact of novel loci in neurodevelopmental disorders and cancer, thus in a secondary analysis we employed an algorithm which searches protein-interaction data for over-representation of genetic variation within interacting proteins in the AVSD trios [22] (Fig.3).

Including protein altering single nucleotide mutations and CNVs derived from the discovery cohort of 59 AVSD trios, the algorithm identified 86 enriched subnetworks of interacting



Fig 3. An illustration of discovery of protein interaction networks for AVSD and validation by burden testing in separate a replication cohort. Filtered SNPs and CNVs from the AVSD trio probands are mapped to a protein interaction network (represented by grey dots and black lines), and the network is pruned to yield subnetworks (green dots and black lines). The subnetworks represent variants in genes which have verified interactions at the protein level, often constituting a portion of a signaling pathway or enzymatic complex [22]. To validate the disease association of the individual subnetworks discovered in the trios, we performed burden testing for each subnetwork in a replication cohort of 100 AVSD cases originating largely from a separate study of AVSDs [10] along with 533 controls without congenital heart disease.

doi:10.1371/journal.pgen.1005963.g003

proteins containing 2 to 7 genes. By comparison, applying the algorithm to 59 control-trios identified 26 subnetworks of interacting proteins containing only 2 to 4 genes (Fig 4a). Using a procedure where the protein interaction network is randomly permuted, the genes within the AVSD-trio subnetworks were found to be enriched for true protein-protein interactions (median p = 0.01, network permutation procedure), while true protein-protein interactions were not observed within the control trio subnetworks (median p = 1.0, network permutation procedure) [23] (Fig 4a).

To further characterize the protein interaction networks detected, we compared the subnetworks to gene expression in mouse cardiac development (<u>S8 Table</u>). Genes within the AVSDtrio subnetworks were strongly overrepresented during mouse heart development (p = 9e-09, one-tailed hypergeometric test) while genes within the control-trio subnetworks were not (p = 0.34, one-tailed hypergeometric test) (<u>Fig 4b</u>). Thus, mapping genetic variation in the AVSD trios to protein interaction networks identifies 86 enriched subnetworks with deleterious variation in 231 genes that are preferentially expressed during cardiac development, a phenomenon not seen in the control trios without CHD.



Fig 4. Protein interaction networks identify collagen genes putatively associated with atrioventricular septal defects. (a) A network diagram displaying protein-protein interactions between genes. A greater number of networks (n = 86) and genes (n = 231) are detected by protein-interaction analysis in AVSD-trio subnetworks (green nodes) relative to the networks (n = 26) and genes (n = 60) detected in the Control-trio subnetworks (yellow nodes). A median *p*-value from a protein interaction network permutation procedure is reported for each group of subnetworks, which shows an enrichment of true protein-protein interactions in the AVSD-trio subnetworks (p = 0.01) that is not observed in the control trio subnetworks (p = 1). The black lines indicate a protein-protein interactions for each observed gene corresponds to the size of the node. The two collagen genes in the AVSD-trio subnetworks (*COL2A1* and *COL9A1*) are labeled and the nodes highlighted in red. The networks from each group of trios are arranged in a circular layout. (**b**) A Venn diagram comparing genes/variants in the AVSD-trio subnetworks (green circle) and Control-trio subnetworks (yellow circle) to genes expressed during early mouse cardiac development (grey circle). The number of genes in each dataset and their overlap with other datasets is indicated and the *p*-value from a hypergeometric test is reported. A greater proportion of genes in the AVSD-trio subnetworks (10 of 60 genes, p = 9.34). (**c**) The subnetwork comprised of *COL2A1* and *COL9A1* are highlighted in a Manhattan plot of test statistics from the SKAT linear weight test for each of the 86 AVSD-trio subnetworks identifies an elevated burden of variation in a replication cohort of 100 AVSD subjects and 533 control subjects. The *p*-value is plotted for the chromosomal position of each gene within the subnetwork. A significance cutoff of 5.81e-04 was derived from the Bonferroni correction of 86 tests performed.

doi:10.1371/journal.pgen.1005963.g004

Protein Interaction Networks Identified in the AVSD-trios are Validated by Burden Testing in a Replication Cohort

To validate the genetic associations suggested by the discovered AVSD-trio subnetworks, we assembled a separate replication cohort of 100 singleton individuals (S2 Table) and performed burden testing for each of the 86 protein networks (Fig.3). After Bonferroni correction for multiple hypothesis testing, a single subnetwork from the AVSD trios composed of a pair of interacting collagen genes (*COL2A1*, *COL9A1*) (Fig 4a and 4c) displayed an elevated burden of rare coding variation in 100 affected individuals with AVSD compared to 533 controls without congenital heart disease (p = 8.37e-08, SKAT linear weighted test) (Fig.4c). Interestingly, the two genes *COL2A1* and *COL9A1* have evolutionarily conserved roles in cardiac development in both zebrafish and mouse [24–26]. One mouse knockout allele of *Col2a1* displays cardiac valve abnormalities [27], and mutations in both genes are associated with Stickler syndrome where 46% of patients are affected with congenital dysfunction of the mitral valve [28]. Together the functional data on *COL2A1* and *COL9A1* accompanied by the identification of these genes with two separate methodologies in two separate cohorts of AVSD patients, support a potential association with other congenital structural malformations of cardiac valve tissue such as AVSDs.

A Rare-Disease Inheritance Model Identifies Known CHD Loci in 23% of Trio Probands

Outside of novel genes identified by developmental coexpression and protein interaction analyses, we wished to examine the impact of genes known to play a role in cardiovascular development or CHD in our cohort. Interestingly, *de novo* single nucleotide mutations in genes previously associated with AVSD (*NKX2-5*, *GATA4*, *GATA6*, *EVC*, *CRELD1*, *NR2F2*) were absent and we did not detect genes with recurrent *de novo* mutations.

In an effort to categorize and catalog variation at known CHD loci within the 710 variants in 399 genes identified by the rare-disease inheritance model in the 59 AVSD-trios (S4 Table), we assembled a predefined group of 756 loci associated with any human or mouse cardiac malformation (S5 Table). Among the genes identified by the rare-disease inheritance model in the 59-AVSD probands (S4 Table), we observed inherited and de novo variation in 16 genes (Table 1 and S7 Table) [29]. Four of the affected probands displayed variation in more than one gene. In a set of 59 control trios without congenital heart disease we applied the identical variant calling pipeline and rare-disease inheritance model. Comparing the number of AVSD probands with inherited mutations in the identified 16 genes to unaffected controls, we observed 16 mutations in the AVSD-cases and only a single variant in controls (p = 9.60e-06, Fisher's exact test), and additional simulations confirmed an unusual distribution of mutations in the AVSD-cases compared to controls was unlikely to be a chance occurrence (p = 1.23e-06, Monte Carlo simulation). As an additional negative control we compared mutations in a list of 43 genes associated with congenital ocular malformations between the 59 AVSD cases and 59 controls; there was only a single inherited variant among the AVSD trios and none within the control trios (p = 1.0, Fisher's exact test). Together these results suggest a preponderance of de novo and inherited variation in genes associated with human or mouse cardiac malformations detected in the AVSD trios which greatly exceeded similar variation in control trios.

Within the discovery cohort of 59 trios we also assessed structural variation by array CGH and read-depth analysis from exome studies. One patient was observed to have a 3.7 Mb *de novo* deletion at 8p23.1 encompassing 43 genes including *GATA4* (Table 2a). An additional paternally inherited duplication at chr22:21,989,140–23,627,391 partially overlapping the congenital heart disease-associated 22q11.2 duplication syndrome region was also detected [30].

Gene	Proband	Inheritance	Amino Acid Changes	Prior Evidence
ADAM17	3721	compound heterozygous	pound heterozygous p.L97R & p.S90L	
RYR1	1131	compound heterozygous	p.E30V & p.D4500H	Mouse Phenotype [42]
CHRD	7952	compound heterozygous	p.A159V & p.T362I	Mouse Phenotype [47]
PTPRJ	3731	compound heterozygous	p.V372I & p.E173K	Mouse Phenotype [46]
IFT140	8522	compound heterozygous	p.E1065K & p.V108M	Mouse Phenotype [41]
ATE1	28	homozygous rare	p.L71F	Mouse Phenotype [45]
NOTCH1	2933	compound heterozygous	p.V2285I & p.R509H	Clinical Association
NSD1	155	compound heterozygous	p.A933P & p.R361S	Clinical Association
ZFPM2	2020	compound heterozygous	p.E30G & p.S388G	Clinical Association
МҮН6	2023	compound heterozygous	p.A936V & p.D208N	Clinical Association
VCAN	54	compound heterozygous	p.V81I & p.E199G	Clinical Association
МҮН6	54	homozygous rare	p.E1295Q	Clinical Association
SRCAP	7952	compound heterozygous	p.S377N & p.E1697G	Syndrome with CHD [49]
KMT2D	2002	compound heterozygous	p.R652H & p.R83Q	Syndrome with CHD [52]
NOTCH2	155	compound heterozygous	p.L2408H & p.D1327G	Syndrome with CHD [51]
BBS2	135	compound heterozygous	p.G688R & p.L62fs	Syndrome with CHD [50]
BBS2	3731	compound heterozygous	p.A504V & p.K231E	Syndrome with CHD [50]
EHMT1	2028	deNovo	p.R260Q	Syndrome with CHD [49]

Table 1. AVSD trio probands display de novo and inherited variation in genes related to human and mouse cardiac malformations.

Among 710 variants in 399 genes displaying *de novo*, homozygous, compound heterozygous inheritance in the AVSD-trio probands we observed 16 genes associated with human or mouse cardiac malformations. Additional variant annotations are found in <u>S4 Table</u>.

doi:10.1371/journal.pgen.1005963.t001

Additional CNVs with a previous association to CHD were identified in the singleton subjects (<u>Table 2b</u>) [<u>31</u>]. CNVs in these regions were absent from the trio and singleton controls. Thus, including inherited variants, *de novo* mutations, and structural variation, rare deleterious variants or CNVs in genes with strong prior evidence for association with congenital heart disease were observed in 14 of 59 or 23% of affected trios.

Table 2. Discovered CNVs associated with AVSDs and other cardiac malformations.

a. Trios						
Coordinates (hg19)	Length	Type & Inheritance	Genes	Additional Information		
22:21989140– 23627391	1,638,251	Duplication, paternally inherited	28 genes including MAPK1	Overlaps with 22q11.2 duplication region [30]		
8:8175651-11853818	3,678,167	Deletion, de novo	43 genes including GATA4	Canonical CHD gene [29]		
b. Singletons						
Coordinates (hg19)	Length	Туре	Genes	Additional Information		
Y:119064-2562294	2,443,230	Deletion	13 genes, pseudoautosomal region	Previously associated with conotruncal CHD [31]		
1:145645983– 145763756	117,773	Duplication	5 genes including CD160 PDZK1	1q21.1 risk locus [<u>29]</u>		
4:5692835–5800635	107,800	Deletion	EVC, EVC2	Causal gene in CHD associated Ellis-van-Creveld syndrome		
8:10755273-11929256	1,173,983	Deletion	14 genes including GATA4	Canonical CHD gene [29]		

Reported CNVs were larger than 100,000 bp, and detected by at least two of three independent methodologies (arrayCGH, consensus read depth based analysis; Conifer & XHMM, or a TaqMan CNV assay) and overlapped known CHD loci and were absent from controls. Coordinates refer to the hg19 build of the human genome reference.

doi:10.1371/journal.pgen.1005963.t002

Discussion

In this study combining both exome-sequencing and array-CGH for a single specific cardiac malformation, we observed *de novo* and inherited variation in 19 genes associated with human disease, syndromic loci, and genes implicated in cardiac development by mouse knockout. In the absence of recurrent *de novo* events in a moderately sized cohort of 159 affected individuals, we applied an array of analytical techniques to look for both *de novo* and inherited variation associated with AVSDs.

Coexpression Networks Suggest a Novel CHD Locus

When combined with inheritance analysis, a gene-coexpression network derived from mouse development allowed us to identify a previously unrecognized role for the transcriptional repressor NR1D2 in cardiac development and human disease. Our experimental studies suggested that the observed p.R175W mutation impacts the transcriptional activity of murine Nr1d2, and we observed previously unrecognized cardiac malformations in the $Nr1d2^{tm1-Dgen}$ knockout mouse. Yet within a cohort of 159 affected individuals, there was only a single patient with a *de novo* mutation in NR1D2, which highlights the underlying genetic heterogeneity of CHD and the utility of applying orthogonal datasets to pinpoint causal variation.

The -/- animals for the $Nr1d2^{tm1-Dgen}$ allele display incomplete penetrance; a majority of animals do not display cardiovascular malformations and develop normally to adulthood displaying phenotypes related to circadian rhythm and abnormal lipid metabolism [20]. Nr1d2 may retain multiple roles in modulating transcription but is most clearly described as a transcriptional repressor, therefore a mutation in the DNA binding domain might impact Nr1d2 to binding to target sequences resulting in a "de-repression" transcriptional targets of Nr1d2. A knockout allele such as $Nr1d2^{tm1-Dgen}$ might similarly "de-repress" targets of Nr1d2. From the standpoint of transcriptional repression, increased transcription of the p.R175W mutant observed *in vitro* may represent a decrease in transcriptional repression relative to the wildtype protein, and as such could be entirely consistent with the phenotype of a mouse knockout allele. Interestingly, NR1D2 is a well-characterized component of the molecular clock, and further studies would be necessary to investigate NR1D2 as a link between the molecular clock and timing of cardiac development.

Chromatin remodeling factors have recently been implicated as primary and secondary causal factors in CHD [6,32], and both newly discovered factors *NR1D2* and *NR2F2* may play integrated roles in chromatin remodeling during cardiac development. The key histone deace-tylase HDAC1 is directly activated by NR1D2 binding and indirectly activated by NR2F2 via PROX1 [33–35]. Additionally NR1D2 may function upstream of NR2F2, modulating the auto-regulatory activity of NR2F2 via HDAC1 and the glucocorticoid receptor GR complex [36,37]. Further experiments are necessary to delineate the tissue localization, timing, expression, and functional roles of these two transcription factors and their role in chromatin modulation and transcriptional regulation during cardiac development.

Exploring Oligogenic Inheritance with Protein Interaction Networks

Within a complex cellular or tissue signaling pathway, capturing the genetic variation in one or more interacting proteins has yielded novel candidate genes in cancer and neurodevelopmental disorders [22,38]. Adapting a tool designed to search protein interaction networks in cancer, we identified a small number of variants in genes within the AVSD trios, two of which (*COL2A1* and *COL9A1*) were subsequently validated in burden testing of a separate replication cohort of 100 individuals at a statistically significant threshold. Independent experimental data implicates these genes in the development of the cardiac valve structures, and links these genes

to a genetic syndrome that includes abnormalities of the cardiac valves among a host of other phenotypes. Though our observations are not firmly conclusive of a causal role for *COL2A1* and *COL9A1* in the pathogenesis of AVSDs, they are supportive of such a role, and we believe, consistent with the idea that network-based approaches may be fruitfully applied to gene discovery in CHD phenotypes.

Absence of Recurrent Mutations in Canonical AVSD Genes

Surprisingly, with the exception of a deletion encompassing *GATA4* seen in one trio subject and one singleton subject (<u>Table 2</u>), we did not discover *de novo* coding mutations or gene dosage alterations within the 59 trios in canonical AVSD genes (*NKX2-5, EVC, CRELD1, GATA6*) or at the newly discovered CHD risk locus *NR2F2*. This finding is consistent with studies of CHD examining candidate genes [<u>39</u>] and exome sequencing where protein-altering variants in any single gene are reported in no more than 1–4% of patients [<u>7</u>]. The absence of recurrent *de novo* variants in a cohort of 59 patients with AVSD is in striking contrast with other cardiac conditions such as long QT syndrome where pathogenic coding variation in only 5 genes accounts for disease in 70% patients [<u>40</u>]. We hypothesize that the absence of *de novo* variation observed at canonical loci in a cohort of this size reflects the complexity of CHD genetics and highlights the utility of considering alternative inheritance patterns to detect disease-associated variation.

Rare Inheritance Models Identify Genes Associated with CHD and Cardiac Development

Among a list of 756 genes with either a clinical or experimental association to cardiac malformations (<u>S5 Table</u>) we observed rare inherited variants in the AVSD trios that were not seen in control trios [29]. In genes clinically associated with CHD we observed compound heterozygous variants inherited *in trans* in *ZFPM2*, *NSD1*, *NOTCH1*, *VCAN*, and *MYH6* and rare homozygous variants in *MYH6*. Variation was observed in 9 additional genes including compound heterozygous variants in *ADAM17*, *CHRD*, *IFT140*, *PTPRJ*, and *RYR1* and rare homozygous variants in *ATE1*, and the presence of heart defects in mouse knockout models for these loci supports their association with human cardiac malformations.

In an independent forward mutagenesis screen *Ift140* causes AVSD among a variety of congenital malformations [41]. The calcium channel *RYR1* is associated with skeletal myopathies and malignant hyperthermia, but primum atrial septal defects in one mouse allele suggest a role in early cardiac development [42]. The metalloproteinase *ADAM17* may link *NOTCH1* signaling in cardiac valve development to the left-right patterning of the heart [43,44]. Individual knockout alleles of *CHRD*, *PTPRJ*, and *ATE1* each show defects in heart development recapitulating different human malformations [45–47]. Importantly, we observed rare inherited variation in genes with experimental or clinical evidence for a role in cardiac development and CHD within the AVSD trios, but rare inherited variation in these same genes was largely absent from the control-trios.

Syndromic Alleles in Nonsyndromic Patients

Despite excluding syndromic features and developmental delay from our patients at the time of recruitment, we observed inherited and *de novo* variation in genes causing syndromic disease that include heart malformations. A *de novo* mutation was detected in an unknown protein domain of *EHMT1* the causal gene in Kleefstra syndrome, compound heterozygous variants inherited in trans were observed in *SRCAP* which was recently associated with Floating-Harbor syndrome, two individuals showed compound heterozygous mutations in *BBS2*

which causes Bardet-Biedel syndrome, a fifth individual displayed compound heterozygous mutations in *NOTCH2* which causes Alagille syndrome, and a sixth individual displayed a compound heterozygous mutation in *KMT2D* the locus implicated in Kabuki syndrome. Each of these multi-organ syndromes is frequently accompanied by AVSD or another related form of congenital heart disease [48–52].

Within a single locus associated with a genetic syndrome, different alleles may vary in their expressivity. We hypothesize that these variants represent hypomorphic alleles of syndromic genes, where the patients affected present only one aspect of the phenotype associated with the syndrome, in this case a phenocopy of a syndromic associated cardiovascular malformation [53]. Indeed on secondary followup, none of the included probands with *EHMT1*, *KMT2D*, *SRCAP*, or *NOTCH2* variants displayed other characteristics of their associated syndromes, while the patients with *BBS2* mutations were not available for review (additional phenotypic information on patients carrying syndromic alleles is detailed in the <u>S1 Text</u>). Supporting the possibility of hypomorphic alleles, there was a striking absence of *de novo* or inherited variants in these syndromic genes within the control trios suggestive that the discovered variants may confer risk for AVSD.

Limitations and Conclusions

These findings have limitations. Although we excluded patients with a family history of cardiac malformations, in an earlier era of surgical care the parents of the study participants would have been less likely to survive to reproductive age [54]. In our study the parents received only a questionnaire and did not receive screening echocardiogram, thus we cannot rule out that a parent in an included trio may have a *forme fruste* of an AVSD-related malformation such as a cleft mitral valve or ostium primum ASD. Additionally, there is emerging evidence that maternal risk factors (both genetic and environmental) which confer risk for CHD that were not considered in our study design [55–57].

Genetic studies of CHD are challenged by the fact that specific individual malformations are quite rare (AVSD is approximately 0.3 per 10,000 live births), and that any substantial group of patients with a single malformation will contain some population stratification. The unexplored role of non-coding gene regulatory variation in congenital heart disease is not surveyed by our exome-sequencing approach [58,59]. The power of SKAT tests are likely limited by a small cohort size, the heterogeneous genetic backgrounds of the case and control populations, the differences in exome sequence capture and sequencing chemistries employed, the absence of an inheritance model, and most importantly the underlying complex oligogenic architecture of cardiac malformations [12,60,61]. Finally there are no statistical models that account for ethnicity in models of rare-variant transmission, therefore the influence of population stratification or ethnicity upon our rare-inheritance model of disease is not known.

Overall our analysis suggests locus heterogeneity in the pathophysiology of a single cardiac developmental malformation. We observed recurrent variation within three genes (*GATA4*, *MYH6*, and *BBS2*) in only 6 of 59 trios. Including inherited, *de novo*, and discovered loci, 32% of trios displayed one or more putatively contributory mutations in the 19 genes identified. Supported by both experimental and clinical evidence, we suggest that inherited rare variants with a moderate effect size across multiple loci may impact the risk of congenital heart disease in addition to *de novo* variation. Taken together our catalog of 19 loci with experimental evidence for disease among 159 patients is consistent with the long-hypothesized oligogenic inheritance of congenital heart disease [12].

Materials and Methods

Ethics Statement, Case Description, and Patient Cohorts

The guidelines of the Declaration of Helsinki were followed and the study was approved by the institutional review board of Stanford University (IRB-23637, IRB-23572) along with each institution from which participants were recruited. Written informed consent was obtained from each participant. Trios or single patients with an AVSD and situs solitus were recruited. We excluded patients with other major congenital malformations, developmental delay, or other types of CHD (excluding patent ductus arteriosus or secundum atrial septal defect). All participants were assessed clinically by pediatric cardiologists or clinical geneticists to exclude other syndromes associated with AVSD or CHD. Participants were obtained from Seattle Children's Hospital [62], the Pediatric Cardiac Genomics Consortium, the University of Iowa, and reanalyzed from a published study of AVSDs drawn from patients at the University of Toronto [7] (S2 Table).

Control Cohorts

Exome sequences for 531 control subjects of Caucasian and African American descent without CHD were derived from the atherosclerosis risk in communities (ARIC) consortium warehoused at dbGAP, and from local recruitment efforts. An additional 59 control trios (healthy parents with a healthy child) were obtained from the Simons Foundation. Raw sequence data in the form of bam or fastq files was re-aligned and re-analyzed with the below-described bio-informatic pipelines.

Exome Sequencing and CNV Detection

DNA was isolated by standard techniques from either whole blood, saliva samples, or immortalized lymphoblasts. Exome sequencing was performed for all complete trios and single affected individuals at two academic centers (University of Washington or Yale University) and two commercial sequencing providers with the SeqCap EZ Human Exome Library v2.0 (Roche NimbleGen, Madison, Wisconsin, USA), SureSelectXT Human All Exon V4 (Agilent Technologies Inc., Santa Clara, California, USA), or a proprietary capture library based on the Agilent SureSelectXT Human All Exon V5 platform (Personalis, Corp, Menlo Park, California, USA). Paired-end sequencing was performed on Illumina HiSeq 2500 machines with 75-, 100-, or 150-bp read lengths in all but two trios, which were sequenced with 33 bp paired end reads. For all included samples Median Ts/Tv was 3.10 while coverage depth was 43.8x (<u>S3 Table</u>). Exome sequencing on single individuals from the Toronto cohort was performed as described [7], and for subset of unrelated individuals raw fastq files were obtained and reanalyzed via the below described bioinformatics platform for the purposes of reanalysis.

AVSD trios and the singleton/replication patients not originating from Toronto were assayed for CNVs by array comparative genomic hybridization (CGH) by using a custom chip described previously or the SurePrint G3 Human CGH Microarray Kit, 8x60K (Agilent Technologies Inc., Santa Clara, California, USA) with described protocols [31,62]. When a CNV was detected in an affected participant, a dye-swap experiment was repeated to ensure reproducibility and the parents were then assayed if available to determine inheritance status. For detection of smaller variation, two exome CNV detection assays using read depth data were also employed for all participants [63,64]. When there was agreement in a structural variant call between at least two calling methods for any variant of interest, we employed an orthogonal CNV genotyping assay when DNA was available (for all of the AVSD trios and the singleton/ replication patients not originating from Toronto). The CNV genotyping assay was carried out

with 10 ng of DNA per manufacturer instructions against the RNaseP CNV reference assay (Life Technologies, Carlsbad, California, USA). The assay was run on a ViiA 7 Real Time PCR System (Life Technologies, Carlsbad, California, USA) for 40 cycles under standard reaction conditions, and CNV genotypes were called with the copycaller software [65].

Genotyping Pipeline

Two genotyping pipelines were tested. A rapid and sensitive commercial software package, rtgcore version 3.3.2 was applied to the raw exome sequence data for mapping, pedigree-aware variant calling, and genotype filtration (Real Time Genomics Inc., Hamilton, New Zealand) [66] to the UC Santa Cruz human genome reference sequence (hg19) (<u>S1 Script</u>). A second pipeline based on the HugeSeq BWA/GATK HaplotypeCaller pipeline was also employed for purposes of comparison.

To determine the accuracy of our genotyping pipeline we re-genotyped available exome data from NA12878 for comparison to a recently described gold standard dataset of 24,734 variants from this individual [14]. This set of variants from the consensus standard dataset was limited to the exome capture region of the nextera kit all human exome v2 (Illumina Corp, San Diego, California, USA) and regions of suspicious variant quality [67] were excluded to yield a true positive dataset of 24,734 true positive variants from NA12878. The vcfeval tool from RTG was used for all comparisons of vcf files as it robustly handles the different possible textual representations of insertions, deletions, and substitutions that may be produced by different genotyping algorithms. Using raw fastq files generated by the Garvan Institute, comparing the unfiltered output of both pipelines the two algorithms both called 19,566 variants (79.9%) of the NA12878 true positive dataset (TP) in common. RTG called an additional 1,469 TP, compared to BWA/GATK which only called an additional 198 unique TP variants. Overall RTG displayed a greater unfiltered sensitivity at 84.5% compared to 79.9% for BWA/GATK; therefore we selected the RTG pipeline for further analysis of the cohort (S1 Table).

To classify variants we selected an AVR score of 0.5 to balance a sensitivity of 99.4% and positive predictive value of 90.3% for the purposes of variant discovery. Alignments for all variants of interest were manually inspected with the IGV software, and all *de novo* coding variants and any insertion/deletion of interest was confirmed by direct Sanger sequencing of PCR amplicons, or alternately clonal Sanger sequencing of 12 colonies from cloned PCR amplicons. For burden testing, pooled simultaneous variant calling for all included cases and controls was performed with the RTG population caller and variants filtered for a read depth of 8 and AVR score of 0.5. Analyses were limited to the regions of intersection of the bed files of the exome capture kits obtained from the respective manufacturers. Known artifactual variants arising from exome sequencing were removed at the time of variant filtration [67,68].

Statistical Analysis

All statistical analyses employed the R language for statistical computing version 3.1 unless otherwise specified. For population analyses we selected 8,940 snps from the Affymetrix Genome-Wide Human SNP Array 6.0 (Affymetrix, Santa Clara, California, USA) common to the five exome capture protocols employed in the study. All included probands were re-geno-typed for this limited set of variants, and combined with individual level data from 1032 diverse samples of known ethnicity from the 1000 genomes project. The MDS and kinship modules from the KING software were used to estimate ethnic background and five individuals displaying cryptic familial relationships from the Toronto cohort among the singletons were identified and excluded from further analysis [69]. A multinomial linear model was built for each population and the presence of admixture from the populations of the 1032 known samples, and used

to infer ethnic background and the presence of admixture in the included probands. Self-reported ethnicity was available for 547 individuals, which was 97.6% concordant with predicted ethnicity.

Protein altering variants were sorted for minor allele frequency less than 0.03 and prioritized by inheritance patterns consistent with rare disease (*de novo*, rare homozygous, and compound heterozygous) using the trioTools module from the STMP package [16]. Protein-altering variants from the 59 trios were selected, haplotypes constructed, and variants phased with the PLINKseq software package [70]. Imputation was not performed thus all variants analyzed originated from the primary genotyping pipeline.

After sorting for inheritance consistent with rare disease (*de novo*, rare homozygous, and compound heterozygous), variants were collated and assembled into lists. A blacklist of genes with low prior likelihood of causality was excluded from further analysis, which included genes with a residual variation intolerance score greater than 90 and genes with copy number polymorphisms [68,71,72].

For statistical comparisons between the groups of trios, the number of individuals with rare inheritance (*de novo*, rare homozygous, or compound heterozygous) in these genes in the 59 trios and 59 controls was counted and compared with a Fisher's Exact test. As the underlying distribution combining *de novo*, rare homozygous, and compound heterozygous inheritance into a unified "rare inheritance" model is not readily estimated from available genotyping data, and because variant detection (particularly *de novo* variation) may vary systematically with the technical aspects of sequencing (variant calling algorithm, read depth, exome capture platform, and sequencing chemistry) we estimated empiric *p*-values by Monte Carlo simulation. To simulate an underlying distribution we performed permutations drawing 16 genes randomly from a list of 18,495 protein-coding genes, the rare inheritance events for each random list counted within 59 cases and 59 controls by individual, and a Fisher's exact test applied.

The *p*-value was estimated with the formula (r+1)/(n+1) where *n* is the number of simulated replicate samples and r is the number of test statistics exceeding the calculated test statistic from the observed data (*p* = 9.60e-06). This simulation procedure suggested an empiric *p*-value of 1.23e-06, a similar order of magnitude as the *p*-value calculated from the observed data.

Gene modules or subnetworks identified by protein interaction networks in the HotNet2 algorithm (see below), were subjected to burden testing with the SKAT linear weight test and a Bonferroni correction for multiple hypothesis testing employed [73,74]. The first four principal components from a principal components analysis of all variants were used as covariates for burden testing. Within the SKAT algorithm variant weighting was derived from a beta density function (p_i , 1, 25) where p_i is the minor allele frequency. For variant weighting, minor allele frequencies were derived from the 61,428 individuals in the multiethnic EXaC dataset (version 0.2 <u>http://exac.broadinstitute.org/</u>), while for variants not observed or reported in the EXaC dataset minor allele frequencies were calculated from frequency observed among the genotyped individuals. The quantile-quantile plots for the SKAT linear weight test of 86 subnetworks (inclusive of combinations of 231 single genes—see below) suggest that the test-statistics derived are controlled for ethnicity or other systematic differences between cases and controls such as coverage (<u>S2 Fig</u>).

Co-expression Network Analysis from Mouse Organ Development

Unique tag-count data from 74 SAGE libraries representing 22 tissues constructed as a part of the mouse atlas of gene expression project were downloaded from <u>www.mouseatlas.org</u> [17]. SAGE tags were mapped with the Burrows-Wheeler aligner to the 115,746 unique mouse RefSeq transcripts downloaded from UC Santa Cruz website (<u>http://genome.ucsc.edu</u>) and tag

data converted to a digital gene expression format constituting a tag counts per transcript using custom Perl scripts. Tags matching pseudogenes were removed. Using the R environment for statistical computing, tag counts by transcript were normalized by library size with the EdgeR package and collapsed to gene by connectivity with the WGCNA package [75,76]. A standard WGCNA workflow for digital gene expression was applied to the normalized and collapsed data for coexpression module construction followed by correlation to the tissue of origin and annotation with the human orthologous gene name when available (S2 Script). To independently assess the predictive capacities of our unsupervised network building procedure, we observed that the developmental expression of 13 well characterized congenital heart disease genes are accurately localized by the network model to their appropriate cardiac tissue, suggesting excellent specificity for detecting developmental cardiac related gene expression (S9 Table). Comparison of variant and gene lists for under- and over-representation were performed with a one-tailed hypergeometric test. Gene expression networks were visualized with the Gephi software package [77]. Because the DMP is a key structure in development of the atrioventricular septum [78] and was not explicitly included in the original set of micro-dissected tissues from the mouse atlas of organ development (www.mouseatlas.org), we added the DMP gene expression data to the cardiac development gene expression set. Among genes expressed in the DMP we selected the most highly expressed 1000 genes across 6 datasets generated from the posterior second heart field including the dorsal mesenchymal protrusion. The microdissection of the posterior heart field was performed in the laboratory of Dr. Moskowitz at embryonic mouse tissue at E9.5, subject to reverse-transcription, amplification, and sequencing by The University of Chicago Genomics Core (GSE75077).

Protein Interaction Networks with HotNet2

Variant data annotated with the STMP package [16] including SNVs and CNVs from the 59 AVSD trio probands and 59 control trio probands was converted and formatted with custom python scripts, and included all protein altering variants with a minor allele frequency in EXaC of 0.02 or less were included in the analysis using four included protein interaction networks. For each of three protein interaction networks (Multinet, IrefIndex9, and HINT [22]) four delta values were derived with the network permutation procedure and applied to identify subnetworks. The resulting subnetworks identified by the HotNet2 algorithm were manually inspected for validity, and the subnetworks from the MultiNet protein interaction networks were chosen for further analysis. For each set of subnetwork sizes ranging from 2 to 10, the HotNet2 algorithm derives a *p*-value from the hypergeometric distribution comparing the observed number of subnetworks of size *n* within in a dataset compared to an expected number of subnetworks. The expected number of subnetworks is derived from a computationally intensive network permutation procedure; in this case the MultiNet protein interaction network was subject to 100 permutations limiting the range of *p*-values to a minimum *p*-value to 0.01 and maximum *p*-value to 1. For the AVSD-trio subnetworks the median *p*-value was 0.01 across the nine subnetwork sizes and four derived delta values, in comparison to the control-trio subnetworks where the median p-value was 1.0; this suggested an enrichment in variants occurring in genes with true protein-protein interactions within the AVSD-trios (rather than randomly occurring simulated protein-protein interactions in the permuted networks). A single output run utilizing the MultiNet protein interaction network applying a delta value of 0.00126036397514 yielded 86 subnetworks containing 231 genes in the AVSD trios, and the 86 subnetworks were subject to burden testing in the singleton cohort (see above). Protein interaction data were processed with custom python and shell scripts for visualization using the Gephi software tool [77].

Gene Lists

Rare inherited variants in the AVSD-trios. The variant calling and filtering process in the 59 AVSD-trios produced a list 710 variants in 399 genes (<u>S4 Table</u>).

The AVSD Coexpression Module. The coexpression module highly expressed in the AV-Canal contains 934 genes including four of six known genes causing AVSDs, including *NKX2-5*, *GATA4*, *GATA6*, and *CRELD1* (<u>S6 Table</u>).

Known CHD genes in human & mouse. To develop a list of 756 genes related to cardiovascular malformations in humans an mice we combined a list of 85 genes previously associated with human cardiac malformations [29] along with a list of 671 mouse genes known to cause cardiac malformations similar to human disease obtained from the mouse genome database (accessed 2/3/15; Mouse Phenotype IDs

10425,10435,3105,10412,6113,285,10422,281,279,10429,2925) (<u>S5 Table</u>).

Cardiac development. Genes from the coexpression modules with correlation to one of the six developing cardiac tissues (atrioventricular canal, branchial arches, heart tube, atria, ventricle, and outflow tracts) were combined with RNAseq data from a key developmental structure in atrioventricular septal morphogenesis, the dorsal mesenchymal protrusion. These data combined yielded an early cardiac development gene set of 3,595 loci (<u>S8 Table</u>) [78].

Ocular malformations. To serve as a list of control genes unrelated to cardiac malformations, we developed a list of genes related to ocular malformations (n = 43) derived from the mouse genome database (accessed 7/15/15; Mouse Phenotype IDs 000516, 0002092) and the literature [79] and excluded genes appearing in the cardiac malformation gene list (S10 Table).

Mouse Experiments

Live breeding pairs of the B6;129P2-Nr1d2^{tm1Dgen}/H mouse line (hereafter referred to as Nr1d2^{tm1-Dgen}) were obtained from the European Mouse Mutant Archive (Munich, Germany). Animals were in housed and cared for in AAALAC accredited facilities under standard conditions with oversight and approval from the Stanford University APLAC committee (protocol APLAC-11334). Euthanasia was carried out under anesthesia with isofluorane following APLAC and AAALAC guidelines using carbon dioxide followed by cervical dislocation. Genotyping was performed by PCR of toe or tail clippings with gel electrophoresis using standard methodology with two primer pairs (CAAGTAACAAGCCTGGGACATAAAG and CTTCGTAGAGGGAGTAATATGACAC yield a 517 bp PCR product from the WT allele; CAAGTAACAAGCCTGGGACATAAAG and GACGAGTTCTTCTGAGGGGATCGATC yield a 757 bp product from the knockout allele). Two pairs of heterozygous founder animals were bred, yielding 17 pups (2 wild type, 7 Nr1d2^{tm1-Dgen} +/-, and 8 Nr1d2^{tm1-Dgen} -/-) which did not deviate obviously from expected Mendelian allelic ratios. Spontaneous death within hours after birth occurred in a single +/- and two -/- animals. The thoracic and abdominal cavities of spontaneously deceased homozygous knockout animals were visually inspected to examine the great vessels and visceral situs which were normal, and hearts dissected out and subject to sectioning and H & E staining by standard techniques. Sectioning of tissue samples was performed following either embedding of frozen sections followed by dehydration and fixation in 10% neutral buffered formalin or alternately fixation in 3% paraformaldehyde followed by alcohol dehydration and paraffin embedding. We performed additional matings of +/- and -/animals, and sacrificed mothers to obtain embryos at e16.5 and e17.5. Visualization was performed by brightfield microscopy on a Nikon 90i Eclipse upright with a DS Fi1 camera with a 20x objective. Both P0 Nr1d2^{tm1-Dgen} -/- animals displayed atrioventricular septal defects, a single -/- animal at e17.5 displayed an inlet ventricular septal defect, and single -/- animal at e16.5

displayed an AVSD. In total, 4 out of 15 -/- hearts assayed displayed a cardiac defect. Spontaneously deceased animals were analyzed, and additionally euthanasia of pregnant female mice was performed to obtain embryonic animals.

Cell Culture Experiments

The crystal structure of the DNA binding domain of NR1D1 (RCSB 1A6Y) was visualized with the PyMOL Molecular Graphics System, Version 1.7.4 Schrödinger, LLC (New York, New York, USA). A construct containing wild-type murine Nr1d2 cDNA construct under control of a CMV promoter in the pCS6 expression vector was obtained from Transomic Technologies Inc. (Huntsville, Alabama, USA), and subject to a site-directed mutagenesis yielding a codon switch at p.R175W which corresponds to the de novo mutation observed in the subject with AVSD (S4a Fig). An Nr1d2 response element vector was constructed cloning 5 tandem repeats of an evolutionarily conserved NR1D2 binding site REV-DR2 [80,81] upstream of a minimal CMV promoter driving GFP expression using the pSF-MinCMV-daGFP vector (Sigma-Aldrich Inc, St. Louis, Missouri, USA) (S4b Fig). Site-directed mutagenesis and cloning were performed by a commercial provider GENEWIZ Inc. (South Plainfield, New Jersey, USA), and sequence verified in our own laboratory. The three vectors were subject to routine endotoxin free preparation. Commercially available primary HUVEC cells obtained from Cell Applications (San Diego, California, USA) were seeded at 80% confluency in black transparent-flat bottom 96 well plates (Greiner, North Carolina, USA) and transfected the following day with 100ng of each vector (response-element, wild-type or p.R175W) using Lipofectamine 3000 (Life Technologies, Grand Island, New York, USA). Twenty-four hours after transfection, GFP fluorescence was measured on a Tecan Infinite M1000-multimode plate reader (Tecan Group Ltd, Mannendorf, Switzerland). We performed three transfection conditions including response-element+Nr1d2-WT, response-element+Nr1d2-P403W and response-element alone in 24 technical replicates. Autofluoresence of untransfected wells were averaged, and subtracted from the response-element alone and the two experimental conditions. The highest and lowest value from each condition were excluded from analysis yielding 22 technical replicates per experimental condition, and 4 technical replicates for the response element alone. Statistical analysis and graphing of the transfection experiments was performed in Prism 6 Graphpad Software (La Jolla, California, USA).

Supporting Information

S1 Text. Supplementary results. Additional phenotypic descriptions of individuals with AVSDs accompanied by mutations in syndromic genes or large CNVs. For AVSD patients displaying large CNVs (<u>Table 2</u>) or mutations in syndromic alleles, we returned to the medical record or referring provider to look for evidence of other syndromic features or developmental delay not ascertained at the time or enrollment. These phenotypic results are discussed herein. As noted in the Discussion, the two patients with compound heterozygous mutations in *BBS2* were unavailable for follow up. (PDF)

S1 Fig. Coexpression modules are correlated with organ development. A heat map displaying correlation between the tissues included from the developmental SAGE expression data on the X-axis and the inferred module eigengene (represented by arbitrarily assigned colors) on the Y-axis. Red and green colors represent correlation and anti-correlation respectively as indicated on the right hand Figure legend. Modules related to atrioventricular canal development are indicated with a black box, the single module containing 934 genes including AVSD-genes

and other genes related to cardiovascular malformations is indicated with the blue arrow and box (<u>S6 Table</u>).

(PDF)

S2 Fig. A quantile-quantile plot of *p***-values derived from the SKAT linear weight test of 86 subnetworks in 100 AVSD cases compared to 533 controls.** The 86 test-statistics produced by the SKAT linear weighted test for the AVSD-trio subnetworks in the replication cohort display a normal distribution, suggesting the comparisons are adequately controlled for ethnicity or other systematic differences between cases and controls such as read depth or exome-capture kits.

(PDF)

S3 Fig. A multiple protein alignment of the DNA binding domains of NR1D1 and NR1D2 shows sequence identity surrounding a conserved arginine residue. The relevant portion of a protein sequence alignment of Q6NSM0 (NR1D2) and P20393 (NR1D1) from UniProt.org (accessed 4/7/15) using standard parameters is shown. The DNA binding domains of the two proteins are delineated by the black text and display 96% sequence identity. The crystal structure of the DNA binding domain of NR1D1 is displayed in Fig 3b. The region of multiple alignment in Fig 3a is highlighted in yellow, and the conserved arginine residue altered in the AVSD patient is highlighted in green.



S4 Fig. Diagrams of nucleic acid reagents for the in vitro cell culture experiments for the discovered p.R175W NR1D2 experiment. (a) multiple alignment of DNA and protein sequences for wild type and mutant versions of the human and mouse of the NR1D2 protein.
(b) Synthetic NR1D2 response element cloned into the XmaI site of pSF-MinCMV-daGFP, containing 5 tandem REV-DR2 response elements (highlighted in bold), separated by random DNA sequence generated by a python script.
(PDF)

S1 Script. Example command line for read mapping and variant calling using the RTG software package.

(PDF)

S2 Script. Example R workflow for weighted gene network coexpression network construction and correlation to a tissue specific expression. (PDF)

S1 Table. Raw sensitivity of RTG is greater than BWA/GATK before filtering. (PDF)

S2 Table. Proband ethnicity and enrollment by site. (PDF)

S3 Table. Sequencing and genotyping parameters of case and control subjects. (PDF)

S4 Table. Complete list of 710 variants in 399 genes from 59 trios with AVSD displaying rare inheritance.

(PDF)

S5 Table. Genes associated with cardiac malformations in human disease or mouse knockout model.

(PDF)

S6 Table. A co-expression module of 934 genes includes *NKX2-5*, *CRELD1*, and *GATA4* is enriched for genes associated with cardiac malformations. (PDF)

S7 Table. Rare variant inheritance patterns identify genes associated with cardiac malformations by prior clinical or experimental evidence. (PDF)

S8 Table. 3595 genes expressed in cardiac development by weighted gene coexpression analysis.

(PDF)

S9 Table. 13 well known cardiac developmental genes are appropriately localized to developmental compartment by unsupervised weighted gene co- expression network analysis of SAGE expression data.

(PDF)

S10 Table. 3595 genes expressed in cardiac development by weighted gene coexpression analaysis.

(PDF)

Acknowledgments

We thank Angela Rogers and Rick Dewey for thoughtful commentary on the manuscript. We thank SAP for funding a portion of the exome sequencing. We are grateful to all of the families at the participating Simons Simplex Collection (SSC) sites, as well as the principal investigators (A. Beaudet, R. Bernier, J. Constantino, E. Cook, E. Fombonne, D. Geschwind, R. Goin-Kochel, E. Hanson, D. Grice, A. Klin, D. Ledbetter, C. Lord, C. Martin, D. Martin, R. Maxim, J. Miles, O. Ousley, K. Pelphrey, B. Peterson, J. Piggot, C. Saulnier, M. State, W. Stone, J. Sutcliffe, C. Walsh, Z. Warren, E. Wijsman). We thank Janice Tracey for assistance with echocardiographic examinations. The contents of this publication are solely the responsibility of the authors and do not necessarily represent the official views of the NHLBI. We wish to posthumously recognize Ron Lauer MD of the University of Iowa for banking patient samples with congenital heart disease for future studies such as this one and to thank five anonymous reviewers for suggestions and critical evaluation of the manuscript. Our co-author Edward J. Lammer passed away unexpectedly during the review of this manuscript; we wish to acknowledge his lifelong contributions to the care and fundamental understanding of children with genetic disorders.

Author Contributions

Conceived and designed the experiments: JRP EAA MP TQ. Performed the experiments: JRP KO KS EJL NM VN RK XHY JDS. Analyzed the data: JRP KO VN. Contributed reagents/materials/analysis tools: JRP DB DW MH SM TS TB IPM DN MB SR FH SG EEE RPL VS. Wrote the paper: JRP SM EAA BDG.

References

- Heron M. Deaths: leading causes for 2010. Natl Vital Stat Rep. 2013; 62: 1–97. Available: <u>http://www.cdc.gov/nchs/fastats/lcod.htm\nhttp://www.ncbi.nlm.nih.gov/pubmed/24364902</u>
- Pradat P, Francannet C, Harris Ja, Robert E. The epidemiology of cardiovascular defects, part I: a study based on data from three large registries of congenital malformations. Pediatr Cardiol. 2003; 24: 195–221. PMID: <u>12632215</u>

- Øyen N, Poulsen G, Boyd Ha, Wohlfahrt J, Jensen PKa, Melbye M. Recurrence of congenital heart defects in families. Circulation. 2009; 120: 295–301. doi: <u>10.1161/CIRCULATIONAHA.109.857987</u> PMID: 19597048
- Jenkins KJ, Correa A, Feinstein Ja, Botto L, Britt AE, Daniels SR, et al. Noninherited risk factors and congenital cardiovascular defects: Current knowledge—A scientific statement from the American Heart Association Council on Cardiovascular Disease in the Young. Circulation. 2007; 115: 2995–3014. PMID: 17519397
- Garg V, Kathiriya IS, Barnes R, Schluterman MK, King IN, Butler Ca, et al. GATA4 mutations cause human congenital heart defects and reveal an interaction with TBX5. Nature. 2003; 424: 443–447. PMID: <u>12845333</u>
- Zaidi S, Choi M, Wakimoto H, Ma L, Jiang J, Overton JD, et al. De novo mutations in histone-modifying genes in congenital heart disease. Nature. 2013; 498: 220–3. doi: <u>10.1038/nature12141</u> PMID: <u>23665959</u>
- Al Turki S, Manickaraj AK, Mercer CL, Gerety SS, Hitz MP, Lindsay S, et al. Rare variants in NR2F2 cause congenital heart defects in humans. Am J Hum Genet. 2014; 94: 574–585. doi: <u>10.1016/j.ajhg.</u> <u>2014.03.007</u> PMID: <u>24702954</u>
- O'Roak BJ, Vives L, Fu W, Egertson JD, Stanaway IB, Phelps IG, et al. Multiplex Tarqeted Sequencing Identifies Recurrently Mutated Genes in Autism Spectrum Disorders. Science (80-:). 2012; 1619: 1619–22.
- Kamp A, Peterson Ma, Svenson KL, Bjork BC, Hentges KE, Rajapaksha TW, et al. Genome-wide identification of mouse congenital heart disease loci. Hum Mol Genet. 2010; 19: 3105–13. doi: <u>10.1093/</u> <u>hmg/ddq211</u> PMID: <u>20511334</u>
- Winston JB, Erlich JM, Green Ca, Aluko A, Kaiser Ka, Takematsu M, et al. Heterogeneity of genetic modifiers ensures normal cardiac development. Circulation. 2010; 121: 1313–21. doi: <u>10.1161/</u> CIRCULATIONAHA.109.887687 PMID: 20212279
- Winston JB, Schulkey CE, Chen I-BD, Regmi SD, Efimova M, Erlich JM, et al. Complex trait analysis of ventricular septal defects caused by Nkx2-5 mutation. Circ Cardiovasc Genet. 2012; 5: 293–300. doi: <u>10.1161/CIRCGENETICS.111.961136</u> PMID: <u>22534315</u>
- Gelb BD. History of Our Understanding of the Causes of Congenital Heart Disease. Circ Cardiovasc Genet. 2015; 8: 529–536. doi: <u>10.1161/CIRCGENETICS.115.001058</u> PMID: <u>26082554</u>
- Nora JJ, Nora aH. Maternal transmission of congenital heart diseases: new recurrence risk Figures and the questions of cytoplasmic inheritance and vulnerability to teratogens. Am J Cardiol. 1987; 59: 459– 63. PMID: <u>3812316</u>
- Zook JM, Chapman B, Wang J, Mittelman D, Hofmann O, Hide W, et al. Integrating human sequence data sets provides a resource of benchmark SNP and indel genotype calls. Nat Biotechnol. 2014; 32: 246–51. doi: 10.1038/nbt.2835 PMID: 24531798
- Exome Aggregation Consortium (ExAC) [Internet]. [cited 12 Mar 2014]. Available: <u>http://exac.</u> broadinstitute.org
- Dewey FE, Grove ME, Priest JR, Waggott D, Batra P, Miller CL, et al. Sequence to Medical Phenotypes: A Framework for Interpretation of Human Whole Genome DNA Sequence Data. Funke B, editor. PLOS Genet. Public Library of Science; 2015; 11: e1005496.
- Siddiqui AS, Khattra J, Delaney AD, Zhao Y, Astell C, Asano J, et al. A mouse atlas of gene expression: large-scale digital gene-expression profiles from precisely defined developing C57BL/6J mouse tissues and cells. Proc Natl Acad Sci U S A. 2005; 102: 18485–18490. PMID: 16352711
- Pravetoni M, Wickman K. Behavioral characterization of mice lacking GIRK/Kir3 channel subunits. Genes Brain Behav. 2008; 7: 523–31. doi: 10.1111/j.1601-183X.2008.00388.x PMID: 18194467
- Bettahi I, Marker CL, Roman MI, Wickman K. Contribution of the Kir3.1 subunit to the muscarinic-gated atrial potassium channel IKACh. J Biol Chem. 2002; 277: 48282–8. PMID: <u>12374786</u>
- Solt LA, Wang Y, Banerjee S, Hughes T, Kojetin DJ, Lundasen T, Shin Y, Liu J, Cameron MD, Noel R, Yoo SH, Takahashi JS, Butler AA, Kamenecka TMBT. Regulation of circadian behaviour and metabolism by REV-ERB-α and REV-ERB-β [Internet]. Nature. 2012. pp. 123–127.
- 21. Mouse Genome Database (MGD) at the Mouse Genome Informatics website, The Jackson Laboratory, Bar Harbor, Maine. [Internet]. [cited 1 Jan 2014]. Available: http://www.informatics.jax.org
- **22.** Leiserson MDM, Vandin F, Wu H-T, Dobson JR, Eldridge JV, Thomas JL, et al. Pan-cancer network analysis identifies combinations of rare somatic mutations across pathways and protein complexes. Nat Genet. 2014; 47: 106–114. doi: 10.1038/ng.3168 PMID: 25501392
- Vandin F, Clay P, Upfal E, Raphael BJ. Discovery of mutated subnetworks associated with clinical data in cancer. Pac Symp Biocomput. 2012; 55–66. PMID: <u>22174262</u>

- 24. DeLaughter DM, Christodoulou DC, Robinson JY, Seidman CE, Baldwin HS, Seidman JG, et al. Spatial transcriptional profile of the chick and mouse endocardial cushions identify novel regulators of endocardial EMT in vitro. J Mol Cell Cardiol. 2013; 59: 196–204. doi: <u>10.1016/j.yjmcc.2013.03.016</u> PMID: <u>23557753</u>
- Chakraborty S, Wirrig EE, Hinton RB, Merrill WH, Spicer DB, Yutzey KE. Twist1 promotes heart valve cell proliferation and extracellular matrix gene expression during development in vivo and is expressed in human diseased aortic valves. Dev Biol. 2010; 347: 167–179. doi: <u>10.1016/j.ydbio.2010.08.021</u> PMID: 20804746
- 26. Fang M, Alfieri CM, Hulin A, Conway SJ, Yutzey KE. Loss of β-Catenin Promotes Chondrogenic Differentiation of Aortic Valve Interstitial Cells. Arterioscler Thromb Vasc Biol. 2014; 34: 2601–8. doi: <u>10.</u> <u>1161/ATVBAHA.114.304579</u> PMID: <u>25341799</u>
- Li SW, Prockop DJ, Helminen H, Fassler R, Lapvetelainen T, Kiraly K, et al. Transgenic mice with targeted inactivation of the Col2a1 gene for collagen II develop a skeleton with membranous and periosteal bone but no endochondral bone. Genes Dev. 1995; 9: 2821–2830. PMID: 7590256
- Liberfarb RM, Goldblatt a. Prevalence of mitral-valve prolapse in the Stickler syndrome. Am J Med Genet. 1986; 24: 387–92. PMID: <u>3728560</u>
- Fahed AC, Gelb BD, Seidman JG, Seidman CE. Genetics of congenital heart disease: the glass half empty. Circ Res. 2013; 112: 707–20. doi: <u>10.1161/CIRCRESAHA.112.300853</u> PMID: <u>23410880</u>
- **30.** Portnoï M-F. Microduplication 22q11.2: a new chromosomal syndrome. Eur J Med Genet. 2009; 52: 88–93. doi: <u>10.1016/j.ejmg.2009.02.008</u> PMID: <u>19254783</u>
- Osoegawa K, Iovannisci DM, Lin B, Parodi C, Schultz K, Shaw GM, et al. Identification of novel candidate gene loci and increased sex chromosome aneuploidy among infants with conotruncal heart defects. Am J Med Genet A. 2014; 164A: 397–406. doi: <u>10.1002/ajmg.a.36291</u> PMID: 24127225
- Guo T, Chung JH, Wang T, McDonald-McGinn DM, Kates WR, Hawuła W, et al. Histone Modifier Genes Alter Conotruncal Heart Phenotypes in 22q11.2 Deletion Syndrome. Am J Hum Genet. Elsevier; 2015; 97: 869–77.
- Lee S, Kang J, Yoo J, Ganesan SK, Cook SC, Aguilar B, et al. Prox1 physically and functionally interacts with COUP-TFII to specify lymphatic endothelial cell fate. Blood. 2009; 113: 1856–9. doi: <u>10.1182/</u> <u>blood-2008-03-145789</u> PMID: <u>18815287</u>
- Wang J, Liu N, Liu Z, Li Y, Song C, Yuan H, et al. The orphan nuclear receptor Rev-erbbeta recruits Tip60 and HDAC1 to regulate apolipoprotein CIII promoter. Biochim Biophys Acta. 2008; 1783: 224– 36. PMID: <u>17996965</u>
- Liu Y, Zhang J-B, Qin Y, Wang W, Wei L, Teng Y, et al. PROX1 promotes hepatocellular carcinoma metastasis by way of up-regulating hypoxia-inducible factor 1α expression and protein stability. Hepatology. 2013; 58: 692–705. doi: <u>10.1002/hep.26398</u> PMID: <u>23505027</u>
- Kang H, Kim YS, Ko J. A novel isoform of human LZIP negatively regulates the transactivation of the glucocorticoid receptor. Mol Endocrinol. 2009; 23: 1746–57. doi: 10.1210/me.2009-0009 PMID: 19779205
- De Martino MU, Bhattachryya N, Alesci S, Ichijo T, Chrousos GP, Kino T. The glucocorticoid receptor and the orphan nuclear receptor chicken ovalbumin upstream promoter-transcription factor II interact with and mutually affect each other's transcriptional activities: implications for intermediary metabolism. Mol Endocrinol. 2004; 18: 820–33. PMID: <u>14739255</u>
- O'Roak BJ, Vives L, Girirajan S, Karakoc E, Krumm N, Coe BP, et al. Sporadic autism exomes reveal a highly interconnected protein network of de novo mutations. Nature. 2012; 485: 246–250. doi: <u>10.1038/</u> <u>nature10989</u> PMID: <u>22495309</u>
- Rauch R, Hofbeck M, Zweier C, Koch A, Zink S, Trautmann U, et al. Comprehensive genotype-phenotype analysis in 230 patients with tetralogy of Fallot. J Med Genet. 2010; 47: 321–331. doi: <u>10.1136/</u> jmg.2009.070391 PMID: <u>19948535</u>
- Lieve KV, Williams L, Daly A, Richard G, Bale S, Macaya D, et al. Results of genetic testing in 855 consecutive unrelated patients referred for long QT syndrome in a clinical laboratory. Genet Test Mol Biomarkers. 2013; 17: 553–61. doi: 10.1089/gtmb.2012.0118 PMID: 23631430
- Li Y, Klena NT, Gabriel GC, Liu X, Kim AJ, Lemke K, et al. Global genetic analysis in mice unveils central role for cilia in congenital heart disease. Nature. 2015; 521: 520–4. doi: <u>10.1038/nature14269</u> PMID: <u>25807483</u>
- Zvaritch E, Depreux F, Kraeva N, Loy RE, Goonasekera SA, Boncompagni S, et al. An Ryr1I4895T mutation abolishes Ca2+ release channel function and delays development in homozygous offspring of a mutant mouse line. Proc Natl Acad Sci U S A. 2007; 104: 18537–42. PMID: <u>18003898</u>
- 43. Wilson CL, Gough PJ, Chang CA, Chan CK, Frey JM, Liu Y, et al. Endothelial deletion of ADAM17 in mice results in defective remodeling of the semilunar valves and cardiac dysfunction in adults. Mech Dev. 130: 272–89. doi: <u>10.1016/j.mod.2013.01.001</u> PMID: <u>23354118</u>

- Boskovski MT, Yuan S, Pedersen NB, Goth CK, Makova S, Clausen H, et al. The heterotaxy gene GALNT11 glycosylates Notch to orchestrate cilia type and laterality. Nature. 2013; 504: 456–9. doi: <u>10.</u> <u>1038/nature12723</u> PMID: <u>24226769</u>
- Kwon YT, Kashina AS, Davydov IV, Hu R-G, An JY, Seo JW, et al. An essential role of N-terminal arginylation in cardiovascular development. Science. 2002; 297: 96–99. PMID: <u>12098698</u>
- Takahashi T, Takahashi K, St John PL, Fleming PA, Tomemori T, Watanabe T, et al. A mutant receptor tyrosine phosphatase, CD148, causes defects in vascular development. Mol Cell Biol. 2003; 23: 1817– 31. PMID: <u>12588999</u>
- Bachiller D, Klingensmith J, Shneyder N, Tran U, Anderson R, Rossant J, et al. The role of chordin/Bmp signals in mammalian pharyngeal development and DiGeorge syndrome. Development. 2003; 130: 3567–78. PMID: <u>12810603</u>
- Stewart DR, Kleefstra T. The chromosome 9q subtelomere deletion syndrome. Am J Med Genet C Semin Med Genet. 2007; 145C: 383–392. PMID: <u>17910072</u>
- Hood RL, Lines Ma, Nikkel SM, Schwartzentruber J, Beaulieu C, Nowaczyk MJM, et al. Mutations in SRCAP, encoding SNF2-related CREBBP activator protein, cause Floating-Harbor syndrome. Am J Hum Genet. 2012; 90: 308–313. doi: 10.1016/j.ajhg.2011.12.001 PMID: 22265015
- Billingsley G, Bin J, Fieggen KJ, Duncan JL, Gerth C, Ogata K, et al. Mutations in chaperonin-like BBS genes are a major contributor to disease development in a multiethnic Bardet-Biedl syndrome patient population. J Med Genet. 2010; 47: 453–63. doi: <u>10.1136/jmg.2009.073205</u> PMID: <u>20472660</u>
- Macgrogan D, Luna-Zurita L, de la Pompa JL. Notch signaling in cardiac valve development and disease. Birth Defects Res Part A—Clin Mol Teratol. 2011; 91: 449–459. doi: <u>10.1002/bdra.20815</u> PMID: <u>21563298</u>
- Hannibal MC, Buckingham KJ, Ng SB, Ming JE, Beck AE, Mcmillin MJ, et al. Spectrum of MLL2 (ALR) mutations in 110 cases of Kabuki syndrome. Am J Med Genet Part A. 2011; 155: 1511–1516.
- Wu N, Ming X, Xiao J, Wu Z, Chen X, Shinawi M, et al. TBX6 null variants and a common hypomorphic allele in congenital scoliosis. N Engl J Med. 2015; 372: 341–50. doi: <u>10.1056/NEJMoa1406829</u> PMID: <u>25564734</u>
- Clapp SK, Perry BL, Farooki ZQ, Jackson WL, Karpawich PP, Hakimi M, et al. Surgical and medical results of complete atrioventricular canal: a ten year review. Am J Cardiol. 1987; 59: 454–8. PMID: 3812315
- 55. Priest JR, Yang W, Reaven G, Knowles JW, Shaw GM. Maternal Midpregnancy Glucose Levels and Risk of Congenital Heart Disease in Offspring. JAMA Pediatr. 2015; 1–5.
- Mitchell LE, Agopian AJ, Bhalla A, Glessner JT, Kim CE, Swartz MD, et al. Genome-wide association study of maternal and inherited effects on left-sided cardiac malformations. Hum Mol Genet. 2015; 24: 265–73. doi: 10.1093/hmg/ddu420 PMID: 25138779
- Schulkey CE, Regmi SD, Magnan Ra, Danzo MT, Luther H, Hutchinson AK, et al. The maternal-ageassociated risk of congenital heart disease is modifiable. Nature. 2015; 520: 230–233. doi: <u>10.1038/</u> nature14361 PMID: 25830876
- Sanchez-Castro M, Gordon CT, Petit F, Nord AS, Callier P, Andrieux J, et al. Congenital heart defects in patients with deletions upstream of SOX9. Hum Mutat. 2013; 34: 1628–31. doi: <u>10.1002/humu.22449</u> PMID: <u>24115316</u>
- 59. Smemo S, Campos LC, Moskowitz IP, Krieger JE, Pereira AC, Nobrega Ma. Regulatory variation in a TBX5 enhancer leads to isolated congenital heart disease. Hum Mol Genet. 2012; 21: 3255–63. doi: <u>10.1093/hmg/dds165</u> PMID: <u>22543974</u>
- He Z, O'Roak BJ, Smith JD, Wang G, Hooker S, Santos-Cortez RLP, et al. Rare-variant extensions of the transmission disequilibrium test: Application to autism exome sequence data. Am J Hum Genet. 2014; 94: 33–46. doi: 10.1016/j.ajhg.2013.11.021 PMID: 24360806
- Granados-Riveron JT, Pope M, Bu'lock FA, Thornborough C, Eason J, Setchfield K, et al. Combined mutation screening of NKX2-5, GATA4, and TBX5 in congenital heart disease: multiple heterozygosity and novel mutations. Congenit Heart Dis. 7: 151–9. doi: <u>10.1111/j.1747-0803.2011.00573.x</u> PMID: <u>22011241</u>
- Priest JR, Girirajan S, Vu TH, Olson A, Eichler EE, Portman Ma. Rare copy number variants in isolated sporadic and syndromic atrioventricular septal defects. Am J Med Genet Part A. 2012; 158A: 1279– 1284. doi: <u>10.1002/ajmg.a.35315</u> PMID: <u>22529060</u>
- 63. Krumm N, Sudmant PH, Ko A, O'Roak BJ, Malig M, Coe BP, et al. Copy number variation detection and genotyping from exome sequence data. Genome Res. 2012; 22: 1525–1532. doi: <u>10.1101/gr. 138115.112</u> PMID: <u>22585873</u>

- Fromer M, Moran JL, Chambert K, Banks E, Bergen SE, Ruderfer DM, et al. Discovery and statistical genotyping of copy-number variation from whole-exome sequencing depth. Am J Hum Genet. 2012; 91: 597–607. doi: 10.1016/j.ajhg.2012.08.005 PMID: 23040492
- Mayo P, Hartshorne T, Li K, McMunn-Gibson C, Spencer K, Schnetz-Boutaud N. CNV analysis using TaqMan copy number assays. Current protocols in human genetics / editorial board, Haines Jonathan L. [et al]. 2010. p. Unit2.13.
- Reumers J, De Rijk P, Zhao H, Liekens A, Smeets D, Cleary J, et al. Optimized filtering reduces the error rate in detecting genomic variants by short-read sequencing. Nat Biotechnol. 2011; 30: 61–68. doi: <u>10.1038/nbt.2053</u> PMID: <u>22178994</u>
- Fuentes Fajardo KV, Adams D, Mason CE, Sincan M, Tifft C, Toro C, et al. Detecting false-positive signals in exome sequencing. Hum Mutat. 2012; 33: 609–13. doi: <u>10.1002/humu.22033</u> PMID: <u>22294350</u>
- Lawrence MS, Stojanov P, Polak P, Kryukov GV, Cibulskis K, Sivachenko A, et al. Mutational heterogeneity in cancer and the search for new cancer-associated genes. Nature. 2013; 499: 214–8. doi: <u>10.</u> <u>1038/nature12213</u> PMID: <u>23770567</u>
- Manichaikul A, Mychaleckyj JC, Rich SS, Daly K, Sale M, Chen W-M. Robust relationship inference in genome-wide association studies. Bioinformatics. 2010; 26: 2867–73. doi: <u>10.1093/bioinformatics/</u> <u>btq559</u> PMID: 20926424
- Purcell S, Neale B, Todd-Brown K, Thomas L, Ferreira MAR, Bender D, et al. PLINK: a tool set for whole-genome association and population-based linkage analyses. Am J Hum Genet. 2007; 81: 559– 75. PMID: <u>17701901</u>
- Petrovski S, Wang Q, Heinzen EL, Allen AS, Goldstein DB. Genic intolerance to functional variation and the interpretation of personal genomes. PLoS Genet. Public Library of Science; 2013; 9: e1003709.
- 72. Campbell CD, Sampas N, Tsalenko A, Sudmant PH, Kidd JM, Malig M, et al. Population-genetic properties of differentiated human copy-number polymorphisms. Am J Hum Genet. 2011; 88: 317–32. doi: 10.1016/j.ajhg.2011.02.004 PMID: 21397061
- 73. Clarke GM, Rivas Ma, Morris AP. A flexible approach for the analysis of rare variants allowing for a mixture of effects on binary or quantitative traits. PLoS Genet. 2013; 9: e1003694. doi: <u>10.1371/journal.</u> pgen.1003694 PMID: 23966874
- 74. Ionita-Laza I, Lee S, Makarov V, Buxbaum JD, Lin X. Sequence kernel association tests for the combined effect of rare and common variants. Am J Hum Genet. 2013; 92: 841–53. doi: <u>10.1016/j.ajhg.</u> <u>2013.04.015</u> PMID: <u>23684009</u>
- Langfelder P, Horvath S. WGCNA: an R package for weighted correlation network analysis. BMC Bioinformatics. 2008; 9: 559. doi: <u>10.1186/1471-2105-9-559</u> PMID: <u>19114008</u>
- 76. Miller JA, Cai C, Langfelder P, Geschwind DH, Kurian SM, Salomon DR, et al. Strategies for aggregating gene expression data: the collapseRows R function. BMC Bioinformatics. 2011; 12: 322. doi: <u>10.</u> <u>1186/1471-2105-12-322</u> PMID: <u>21816037</u>
- 77. Bastian M, Heymann S, Jacomy M. Gephi: An Open Source Software for Exploring and Manipulating Networks. Third Int AAAI Conf Weblogs Soc Media. 2009; 361–362.
- Briggs LE, Kakarla J, Wessels A. The pathogenesis of atrial and atrioventricular septal defects with special emphasis on the role of the dorsal mesenchymal protrusion. Differentiation. 2012; 84: 117–30. doi: <u>10.1016/j.diff.2012.05.006</u> PMID: <u>22709652</u>
- Zagozewski JL, Zhang Q, Eisenstat DD. Genetic regulation of vertebrate eye development. Clin Genet. 2014; 86: 453–60. doi: 10.1111/cge.12493 PMID: 25174583
- Kakizawa T, Nishio S, Triqueneaux G, Bertrand S, Rambaud J, Laudet V. Two differentially active alternative promoters control the expression of the zebrafish orphan nuclear receptor gene Rev-erbalpha. J Mol Endocrinol. 2007; 38: 555–68. PMID: <u>17496157</u>
- Harding HP, Lazar MA. The monomer-binding orphan receptor Rev-Erb represses transcription as a dimer on a novel direct repeat. Mol Cell Biol. 1995; 15: 4791–802. PMID: 7651396
- Zhao Q, Khorasanizadeh S, Miyoshi Y, Lazar MA, Rastinejad F. Structural elements of an orphan nuclear receptor-DNA complex. Mol Cell. 1998; 1: 849–861. PMID: <u>9660968</u>
- Anderson RH, Wessels a, Vettukattil JJ. Morphology and Morphogenesis of Atrioventricular Septal Defect With Common Atrioventricular Junction. World J Pediatr Congenit Hear Surg. 2010; 1: 59–67.